Author’s response to reviews

Title: Clinical and Radiological Features and Skeletal Sequelae in Childhood Intra-/juxta-articular Versus Extra-articular Osteoid Osteoma

Authors:

Mi Hyun Song (wwiiw82@snu.ac.kr)
Won Joon Yoo (yoowj@snu.ac.kr)
Tae-Joon Cho (ticho@snu.ac.kr)
Chin Youb Chung (chin@snu.ac.kr)
Moon Seok Park (pmsmed@gmail.com)
Jung-Eun Cheon (cheonje@snu.ac.kr)
In Ho Choi (inhoc@snu.ac.kr)

Version: 2 Date: 27 August 2014

Author’s response to reviews:

Dear Reynaldo Aldea Jr.:
We have revised our paper as the editorial office indicated. Thank you.
1. We stated in the Methods section that a written informed consent for participation in the study was obtained from a parent of the patients (lines 82 and 83).
2. Name of the ethics committee was stated (lines 76 and 77).
3. We provided Acknowledgements section (lines 237 and 238).